Author's response to reviews

**Title:** Protocol for a pragmatic randomised controlled trial to evaluate effects of a brief intervention for emergency department attendees who present with moderate and high levels of non-specific psychological distress: a pilot study.

**Authors:**

Petra Lawrence (petra_lawrence@health.qld.gov.au)
Paul Fulbrook (paul.fulbrook@acu.edu.au)

**Version:** 13  **Date:** 16 September 2015

**Author's response to reviews:** see over
16th September 2015.

Dear Editor

Thank you for your feedback

I have changed the protocol as requested

1: Page 11 regarding ‘Allocation Concealment’. I used the paragraph and used the example that that was suggested

2. Page 14: deleted the word ‘longitudinal’ from the sentence about percentages.

Thank you so much for your time and feedback. I really appreciate it

Petra Lawrence
The Prince Charles Hospital
Brisbane Australia